Author's response to reviews

Title: Spinal cord hemorrhage in a patient with neurosarcoidosis on long-term corticosteroid therapy: case report

Authors:

Benoit BP Pegat (benoit.pegat@gmail.com)
Sophie SD Drapier (sophie.drapier@chu-rennes.fr)
Xavier XM Morandi (xavier.morandi@chu-rennes.fr)
Gilles GE Edan (gilles.edan@chu-rennes.fr)

Version: 5 Date: 18 June 2015

Author's response to reviews: see over
Dear Sir or Madam,

You will find in this cover letter the last revision we made for the case report « Spinal cord hemorrhage in a patient with neurosarcoidosis on long-term corticosteroid therapy. Case report ». Thank you.

Best regards,

Benoit PEGAT

Editor’s comment:

1) “Although the most probably explanation... probable cause of the bleeding.”  
   “So I suggest reformulating the discussion... brain damage”

We reformulate the discussion as you said. We add references supporting the sarcoid vasculitis, the description during the surgery and the previous cerebral hematoma. [line 104 to 113]

2) “The paper of Kreppel... would be better contextualized”

We explained why hematomyelia was rarely diagnose in the meta-analysis of Kreppel, because of the dearth of MRI-based diagnosis prior to 1996. [line 121 to 125].

3) “The conclusion in the abstract needs to be improved”

The conclusion in the abstract was improved adding the immunosuppressive therapy discussion [line 32 to 34]

4) “The authors do not explain how the patient was treated after... included in the conclusions”

We fully described the case after the surgery: drugs, discussion about immunosuppressive therapy.  
Case presentation [line 88 to 91]  
Discussion [line 116 to 120]  
Conclusion [line 138-140]

5) And 6)  
These items were not understandable with “add ?” and “remove ?”

7) “Improve the titles of the figures, 8) Remove L Left”

The titles and the legends of the figures have been rewritten.
9) “English improvement”

We worked with a native English speaker to improve every part of the manuscript.

Moreover, the figure 1 was changed to improve the quality of the pictures.
We add the consent form of the patient.